Author's response to reviews

Title: What are the risk factors of colonoscopic perforation?

Authors:

Varut Lohsiriwat (bolloon@hotmail.com)
Sasithorn Sujarittanakarn (pukky177@yahoo.com)
Thawatchai Akaraviputh (sitak@mahidol.ac.th)
Narong Lertakyamanee (sinla@mahidol.ac.th)
Darin Lohsiriwat (sidls@mahidol.ac.th)
Udom Kachinthorn (siukt@mahidol.ac.th)

Version: 5 Date: 24 September 2009

Author's response to reviews: see over
September 24, 2009

RE: MS: 5877657752626616

Dear the Editor-in-chief of the BMC Gastroenterology

Thank you very much for kindly accepting our manuscript MS: 5877657752626616, entitled "What are the risk factors of colonoscopic perforation?" to be published in the BMC Gastroenterology.

We would like to submit a final version of this manuscript after an amendment of its format. The changes to the manuscript have been described as following: 1) Authors’ affiliations have been re-written, according to the format of BMC Gastroenterology, and 2) The last two rows in table 1 have been adjusted.

Your consideration on this final manuscript would be highly appreciated.

Yours sincerely,
Varut Lohsiriwat MD.
Corresponding author of the manuscript

Department of Surgery, Faculty of Medicine Siriraj Hospital, Mahidol University, Bangkok 10700, Thailand
Phone: +66 (0) 2419 8077
Fax: +66 (0) 2411 5009
Email address: bolloon@hotmail.com